Author's response to reviews

Title: Infratentorial Benign Cystic Meningioma Mimicking A Hemangioblastoma Radiologically & A Pilocytic Astrocytoma Introperatively: Case Report

Authors:

Tan Kheng Guan (devaraj77@gmail.com)
Devaraj Pancharatnam (neri_mala@yahoo.com)
Hari Chandran Thambinayagam (chandran.hari18@gmail.com)
Teoh K.H (devaraj77@gmail.com)
Gnana Kumar P Gnanasuntharam (gnanak_rad@um.edu.my)
Dharmendra Ganesan (drdharmen@gmail.com)

Version: 3 Date: 3 July 2012

Author's response to reviews: see over
To,
The Journal of Medical Case Reports Editorial Team

Title: Infratentorial Benign Cystic Meningioma Mimicking A Hemangioblastoma Radiologically & A Pilocytic Astrocytoma Introperatively: Case Report

Firstly I would like to thank you for considering our manuscript. We have analysed the comments by your reviewers Listed below are the reply to the reviewers report and the corresponding page.

Reviewer #1:

Question 1 .The number of mitosis was 1/hpf

Line 105: and indistinct cytoplasmic borders. Mitoses were seen 1/hpf but there was no necrosis noted. Cytological atypia

Question 2. We have corrected the typographical errors as below (in red)

Line 121: cyst and intratumoral cyst formation have different pathophysiological mechanisms. Peri- tumoral cysts may be

Line126: Pathologically, intra-tumoural cysts were more common in the angiomatous and meningiothelial variants as

Line 141: commonly present as a well circumscribed tumour with a solid mural nodule within a large cystic cavity (5), the

Line 152: case. However the lipid peak on the MRS is non-specific and is also observed in other high grade tumours such as
Question 3: We have added two more high power photomicrograph
(S2011-2430 x20 and S2011-2430x40) in representative of the tumour
morphology.

Reviewer #2:

We unfortunately do not have any intraoperative images during the surgery for this
patient.

We sincerely hope that these changes will be accepted and this manuscript accepted.

Thank you,

Yours sincerely,

…………………………………
Dr. Devaraj Pancharatnam
MBBS(Malaya), MRCSEd, MSurg (Malaya)
Senior Lecturer,
Division of Neurosurgery, Department of Surgery,
Faculty of Medicine, University Malaya,
Kuala Lumpur, Malaysia.